Author's response to reviews

Title: Factors associated with post-traumatic stress disorder and depression amongst internally displaced persons in northern Uganda

Authors:

Bayard Roberts (bayard.roberts@lshtm.ac.uk)
Kaducu Felix Ocaka (fkaducu@yahoo.co.uk)
John Browne (john.browne@lshtm.ac.uk)
Thomas Oyok (oyokthomas@yahoo.co.uk)
Egbert Sondorp (egbert.sondorp@lshtm.ac.uk)

Version: 5 Date: 9 May 2008

Author's response to reviews:

Dear Sir/Madam,

Further to your email of 9 May 2008 (received at 16.36), I have made the following changes in the submitted manuscript:

1. I have removed 'title' from the title page.
2. I have changed 'ABSTRACT' to 'Abstract'.
3. I have put the abbreviations as a paragraph rather than a list.

However, I did not respond to the issue of Figure legends as legends are not required for the figures and it was not entirely clear what change was required.

I apologise for any inconvenience caused to you.

Your sincerely,

Bayard Roberts.